Author's response to reviews

Title: Soft tissue perineurioma of the retroperitoneum in a 63-year-old man, CT and MR imaging findings: a case report

Authors:

Mayumi Yasumoto (mayumihoo@gmail.com)
Yoshiaki Katada (ktdmrad@tmd.ac.jp)
Reiko Matsumoto (rad@saitama-med.jrc.or.jp)
Akiko Adachi (akkia66@mn.em-net.ne.jp)
Koh Kaneko (byoribu@saitama-med.jrc.or.jp)

Version: 7 Date: 15 June 2010

Author's response to reviews:

Dear Editor

I have changed figure files of Figures 1, 2 and 4, as indicated, and all figures were cropped.

All figures were cited in the main body of the manuscript, and yellow highlighting was deleted.